

Received September 13, 2020, accepted October 12, 2020, date of publication November 9, 2020, date of current version December 16, 2020.

Digital Object Identifier 10.1109/ACCESS.2020.3036976

A Study on the Active Methodologies Applied to Teaching and Learning Process in the Computing Area

MARIA IVANILSE CALDERON RIBEIRO^{D1} AND ODETTE MESTRINHO PASSOS²

¹Institute of Computing, Federal University of Amazonas, Manaus 69067-005, Brazil
²Institute of Exact Sciences and Technology, Federal University of Amazonas, Itacoatiara 69103-128, Brazil
Corresponding author: Maria Ivanilse Calderon Ribeiro (ivanilse.calderon@icomp.ufam.edu.br)

This work was supported in part by the Institute of Computing at the Federal University of Amazonas (UFAM), Manaus, AM - Brazil, in part by the Foundation for Research Support of the State of Amazonas (FAPEAM) - POSGRAD 2017 (Resolution 002/2016), in part by the Coordination for the Improvement of Higher Education Personnel - Brazil (CAPES) - Finance Code 001, and in part by the support of Federal Institute of Education, Science and Technology of Rondônia (IFRO)/Campus Porto Velho North Zone.

ABSTRACT Active Methodologies allow an active process in teaching and learning contents, promote responsible student involvement and bring satisfaction and enrichment to educational practices and active learning. Generally, students have learning difficulties in Computer Science courses, as they need to develop computational skills and thinking. The goals of this article is to characterize and analyze the types of Active Methodologies that are being applied in teaching and learning activities in Computer Science. Thus, this investigation was carried out through a Systematic Mapping Study, focusing on the use of the types of methodologies in view of the results achieved. It presents students' perceptions, benefits, and difficulties in adopting these methodologies in the classroom. The results show 6 types of different Active Methodologies used in 35 publications selected, different types of techniques or studies that were used, the publications trend per year, the courses that were worked in analyzed publications, and some benefits and difficulty related to the adoption of Active Methodologies. Regarding to students' perception, we identified different type feelings. Thus, the contributions of this study consist in a research focused on the use of Active Methodologies in a very broad sense, including the perceptions of teachers and students regarding the use of different teaching and learning methodologies. In addition, it shows the specific benefits and possible difficulties experienced in the use of Active Methodologies as teaching strategies. Consequently, some findings from this study may have the potential to support or direct choices of these methodologies in different Computer Science courses.

INDEX TERMS Active methodologies, teaching in computing, learning, students' perception.

I. INTRODUCTION

The old method of teaching in which students were getting used to stay seated on their places, writing down and listening to a lecture of a teacher, have changed [1]. The nature of computer use has changed remarkably in the past fifty years. However, most Computer Science (CS) courses are still often teaching through that old paradigm that is not adequated to deal with modern concerns. Even in the face of the current generation of students and the nature of computing, most computer courses are still teaching in traditional ways [2]. That scenario needs a new conception that can brought a profound pedagogical renewal that requires knowledge and domain of new methodologies [3].

Active methodologies (AM) can support the development of self-competencies and skills. Because, on an increasingly complex society, mere transmission of information no longer characterizes an efficient teaching and learning process [4]. Therefore, AM are teaching strategies centered on the effective participation of students in the construction of the learning process, in a flexible, interconnected and hybrid way [5]. Also, on the new way, the act of learning needs to become a reconstructive process that allows the students to establish different relationships between facts and objects, producing resignifications and reconstructions and contributing to their application in different contexts [6], brings satisfaction and enrichment for both teacher and students [7].

The associate editor coordinating the review of this manuscript and approving it for publication was Chia-Wen Tsai.

Although there are several papers investigating about AM in teaching and learning, there is still a shortage of papers that shows how these AM influence and also can be exploited for support knowledge of teaching and learning in CS courses. Therefore, we have identified the need for a comprehensive research bout influence and support AM in teaching and learning in CS courses.

Systematic Mapping Studies (SMS) provides an overview of a research area, identifying the quantity, the types of research carried out, the results available, in addition to the frequency of publications over time to identify trends [8]. Our aims in this study, is characterizes and analyzes types of AM most applied to teaching and learning activities in CS courses. It also presents the students' perceptions about the AM applied in teaching, some benefits, and difficulties in adopting the different types of AM inside of a classroom.

From an initial selection of 753 publications, we have identified 35 different publications that have used AM to support in teaching and learning in CS courses. From the selected publications, the following AM were the most cited: Gamification (GM); Problem-Based Learning (PBL); Project-Based Learning (ProjBL); Peer instruction (PI); Flipped classroom (FC); and Team-Based Learning (TBL). Also, we have identified the research goal per AM applied, the publications trend per year, the main proposals of the studies mapped, technical or studies used per methodology, the computing area and active methodology applied, the courses that were worked in analyzed publications, and some benefits and difficulty related to the adoption of AM. Regarding to students' perception, we identified the feelings of Satisfaction in learning the content, Motivation to learn content and Feeling of mastery of content.

Besides this introductory section, this paper is structured as follows: Section 2 present related works; Section 3 describes the research method to define search strategy and research questions; Section 4 discusses the results; finally, Section 5 describes limitations and validity of this article, and Section 6 shows some conclusions.

II. RELATED WORKS

In recent years, several studies have reported evaluations and comparisons regarding of use of AM as teaching strategies in computer science e.g., [9]–[11]. The goal of those studies was to demonstrate the use of the AM with a focus on its use and apply them as teaching methods. Those works point out to AM as possible support to improve teaching in the computing and, consequently, to minimize the avoidance of courses in the area.

Raes *et al.* [12] present the results of an experiment on that subject comparing the students' learning experiences in a lecture face-to-face versus virtual students. The results show that although the hybrid virtual classroom is promising flexibility in education as it gives to students the choice for when and where attend the course, it is also the most challenging one to teach and to learn as a remote participant.

Farias & Nunes [13] argue that many studies in computing and education science present tools or environments related to teach in programming that does not reach the expected effect on the students' learning experience. That research also shows some relevant studies that, involving active programming learning for high school and university students, contribute to the construction of an innovative educational scenarios involving active programming learning.

Silva & Oliveira [14] have published an experience report of how robotics fits as an effective instrument from the perspective of AM for education. Those authors point out that robotics in education presents a possible path to be taken and used as a strategy applied to AM for the development and assimilation of knowledge, capable of promoting increasing participation and interest on the part of students, inside and outside school.

Moreno [15] proposes the discussion of evaluating how the principles, values, techniques and agile development processes can be relate to the AM to optimize teaching and learning process in introductory programming subjects with focus, specifically, on students of the high school. Thus, that author seeks to understand how teaching and learning, in introduction to programming, can be optimized by converging the active techniques widely discussed in the literature with those considered agile in software engineering.

The paper of Silva *et al.* [16], by his turn, seems to aim to present an systematic literature review on the use of digital games on teaching of programming for beginners in computing at university level in the last decade in Brazil. The authors argue that teaching programming is part of the basic academic training in CS and related areas.

We have observed that the use of AM in the context of learning computing still faces some challenges, mainly in relation to the attitude of the teacher and the students in relation to the use and effective applicability of the methodologies. In addition, the students' posture in relation to his autonomy for studies or in relation to an active posture for his studies inside or outside classroom, negatively reflect on skills related to computational thinking and learning computing [17], [18].

Resuming, all those previous works show that, besides a positive differential for teaching computing, today the adoption of AM are still a challenge because some teachers have resistance to adopt new teaching techniques. Furthermore, the training of new professionals in the area of technology, as well as in other areas of knowledge in computer programming, is a challenging task for hundreds of scientific researches, proposing several strategies that address from robotics, programming language, and educational games and to pedagogical approaches [19]. Also, that review has concluded that students have their own problems related to autonomy for study.

III. METHODS

For this paper, we have conducted SMS to collect all evidences that fit eligible criteria pre-specified following the recommendations presented by Kitchenham & Charters [20]. Those kinds of studies also help to identify gaps in current research in order to suggest areas for further investigation [21] and by gathering, synthesizing and reviewing study findings [22]. So, this map was undertaken in three phases: planning, conducting, and reporting based on a systematic review protocol. Details of those stages are described in the following sub-sections.

To examine the current use of AM applied in CS courses, our research questions were:

RQ1. Which types of AM are being applied in teaching and learning activities in computing courses?

RQ2. Which are the courses' perceptions about the AM applied in teaching?

RQ3. Which are the benefits and difficulties in relation to the adoption of AM?

A. DATA SOURCE

The main digital libraries that were used to search for primary studies was Scopus, because: (i) It provides Index for publications of most events in the computer and education area, according to Dybá et al. [23]; (ii) It is an important repositories and are widely used for research in the scientific community; (iii) Its databases provide the best results, have strengths in different areas and return papers from more traditionally indexed [24]. In addition, we also manually searched the symposium and conference proceedings and journals in which relevant studies to the computer and education area domain had previously been published: (i) Brazilian Symposium on Informatics in Education (SBIE); (ii) Brazilian Symposium on Games and Digital Entertainment (SBGames); (iii) Computer Workshop at School (WIE); (iv) Computer Education Workshop (WEI); (v) New Technologies in Education Journal (RENOTE); (vi) Journal of Informatics in Education (RBIE); and (vii) International Congress of Educational Informatics (TISE). The period of analysis of the proceedings of the symposium above was from 2010 to 2019. Also, it is important to mention that in the Brazilian Symposium on Games and Digital Entertainment the search was made in all tracks; however, the proceedings of this symposium of the year 2014 was not used, as it is not available at the time of the search.

B. SEARCH STRATEGY

For the construction and refinement of the search string, we have flowed the recommendations of Petersen *et al.* [8]. We have performed the three advise steps, which are: (i) Consultation to the experts for the construction of the mapping protocol; (ii) The refinement and (iii) Test of the search string and selection of the keywords for manually searched.

Furthermore, to facilitate the identification of search string terms, the terms were defined from the Population, Intervention, Comparison, Output (PICO) parameters, by Kitchenham & Charters [20] and the terms related to each parameter, when applicable (see Table 1).

TABLE 1. Terms used to instantiate parameters PICO.

Parameter	Used terms of search
(P) Population: Works published in conferences or journals presenting models and artifacts involving active methodologies	"active methodologies" OR "methodologies" OR "active learning" OR "teaching computer" OR "educational experiences" OR "computer programming skills"
(1) Intervention: Teaching- learning activities of computer science course	OR course OR computing
(C) Comparison: It does not apply, because it is a characterization review	not applicable, is a mapping for characterization
(O) Output: Concepts, definitions, methods, models, applications, discussion of problems	concepts OR approach OR method OR model

(("active methodologies" OR "methodologies" OR "active learning"
OR "teaching computer" OR "educational experiences" OR "computer
programming skills")
AND
(activities OR teaching OR learning OR course OR computing)
AND
(concepts OR approach OR method OR model))

FIGURE 1. Search string used in the SMS.

Moreover, we used the search string in which Boolean OR has been applied to join alternate terms and synonyms in each main part; and Boolean AND has been used to join the three main parts. Figure 1 shows the search string of this work.

C. STUDY SELECTION

During the SMS, only relevant publications to the research question were selected for further analysis. Kitchenham & Charters [20] had suggested the definition of inclusion and exclusion criteria for papers that are returned by the search string. Any paper that did not meet all the inclusion criteria must be deleted. Therefore, we have used the five inclusion criteria to select articles (see TABLE 2).

D. SEARCH RETURNS AND DATA EXTRACTION

The literature search identified 753 publications. After the removal of 105 duplicate papers, applying the 1st filter (selection based on title, keywords and abstract) and 2nd filter

TABLE 2. Inclusion set of criteria.

id	Inclusion criteria
Inc 1	Publication must present topics related to the use of AM in teaching activities in the context of computer and education area.
Inc 2	Publication must present details for applying of AM in teaching activities of course computer science.
Inc 3	Publication must present concepts, approach, method or model in the context of computer and education area.
Inc 4	Articles were written in English and Brazilian Portuguese.
Inc 5	Articles were published between 2010 to 2019.

(complete analysis of the study) to removal out-of-scope and duplicate papers (see Fig. 2), the final number of studies reviewed was reduced to 35 relevant papers that are listed in the (Supplementary data Appendix 1).



FIGURE 2. Shows the results obtained that answers the research.

In this scenario, we can observe that most of the researches selected in this mapping were published in SBGames, i.e., in this symposium the community can find most of publications related to AM teaching strategies in computer science. Also, we can observe SBIE, another symposium also used by the computer community in education. Furthermore, the journals RENOTE and RBIE also have been used.

IV. RESULTS AND ANALYSIS

In this section, we present our results according to the research questions.

A. ACTIVE METHODOLOGIES IN TEACHING OF COMPUTING

RQ1. Which types of AM are being applied in teaching and learning activities in computing courses?

Fig. 3 shows the results of 6 different types of AM most applied in the teaching and learning activities of different CS courses: GM; PBL; ProjBL; PI; FC and TBL.

Between publications analyzed, were applied the TBL and PI, but those types were used not alone in the work. For example, on the publications [10], [25]–[27] the researchers have used more than one AM as teaching strategies in their research. TBL methodology goes beyond covering the content, as it allows the use of course concepts to solve problems. Therefore, on TBL, learning is favored through group interaction; after the questions raised, they are discussed within the groups, the answers are presented to the class, thus revising the main points of the subject [18], [28]. The professional school educators have found TBL particularly attractive because it offers powerful solutions to several major problems they face in teaching [29]. Thus, it is an active learning method developed to help students achieve goals of the course while learning how to function in teams.

FC is a methodology that means that events that have traditionally taken place inside the classroom now take place outside the classroom and vice versa. The use of learning technologies, particularly multimedia, provides new



FIGURE 3. Types of AM are applied. In increasing order: TBL; FC; PI; ProjBL; PBL and GM.

opportunities for students to learn, opportunities that are not possible with other media [30]. Basically, the concept of a flipped class is this: what is traditionally done in class is now done at home, and what is traditionally done as homework is now completed in class. In the flipped model, the time is completely restructured [31].

Furthermore, it is important to mention that the AM applied in conjunction with TBL and PI were the PBL, ProjBL, FC and GM together, i.e., around 14% (5 publications) of set analyzed. PI is collaborative methodology, developed by teacher Eric Mazur of Harvard University. It aims to involve all students during class, promoting activities in which they are encouraged to apply the concepts discussed at that time, while explaining them to their colleagues [18]. Therefore, approach with PI will help them learn, mainly because the students have to play a central role in their own learning with the instructor as their coach [32] [33].

PBL is a methodology emerged in the 70s, through the doctor Howard Barrows, being applied in medical classes and has gained acceptance and is becoming increasingly effective across a variety of course in higher education and an educational method for the teaching of computing is being used in computer science [9]. It is considered to be an educational strategy centered on the students, which helps him in the development of reasoning and communication, essential skills for success in his professional life [34].

Fig. 3 shows the mapping results, PBL, ProjBL, PI and FC types of AM that were used in a few publications. Those methodologies were used just in 17% (7 publications), 9% (3 publications) and 6% (2 publications) respectively of the total set analyzed. Therefore, researches that use the methodology PBL, in general, seems to challenge the students to perform high-level mental tasks, such as analysis, synthesis and evaluation. That method was used totally focused on the health area, but nowadays it has been accepted in the teaching of several areas of knowledge, mainly in Computing, both on elementary and on high school [9].

ProjBL is a methodology that organizes learning around projects, according to definitions found in Project-based learning handbooks for teachers. For this methodology, projects are complex tasks, based on challenging questions or problems, that involve students in design, problem-solving, decision making, or investigative activities; give students the opportunity to work relatively autonomously over extended periods of time; and culminate in realistic products or presentations [35] [36].

Ultimately, around 71% of the studies (25 publications) of set analyzed presented works used GM as methodology to teaching courses or measure the learning. Thus, it is the use of game elements and design for purposes unrelated to games to get people motivated to achieve specific goals [37] [38]. Therefore, gamification can provide an edge in learning delivery when it is designed, developed, and deployed properly. Thus, the effort should not focus solely on points, badges, and leader boards. Results of the change have bilateral naturethey can affect students' results and help them to understand the educational content and create conditions for an effective learning process [39]. We have observed that, on large-scale, concepts of gamification are applied into other areas almost every day [40]. And, on educational context, it is no different, because those researches have linked positive impacts on game experiences in different cases, particularly on player's experiences and interactions during learning courses on computing.

That scenario address that there is a greater production of procedure or technique for teaching in computing applying AM. Thus, we can notice a great concern of the educational community that demands of teachers to prepare dynamic classes and master how to use the different AM to motivate students in addition to producing different materials. It is important, because an increasing number of strategies are gaining prominence in terms of getting students' attention, changing the traditional way of teaching and learning [1], and support the search for learning objects that might help teachers on that task.

We also have analyzed the main objective of the researchers in relation to the use of AM; those research interests present different challenges encountered in the studies for the application of AM on teaching of computation (see Table 3). Those investigated studies, in general, present their objective as a search for different strategies for teaching and learning applying different AM, which results in different solutions for teaching.

This overview of research goal shows the importance of knowing about that topic and the need for developing more work to understand and increase useful tools to improve different means about teaching and learning. It is so because educational field is facing an impasse due to numerous changes in society: it is necessary to evolve and to make everyone learn in a competent and constructive way [41]. Also, the current social demands require much more from teachers in the classroom, not only a new attitude, but new ways of transmitting their knowledge [42].

Table 3, in general, shows concerns related to teaching and learning of content by students, research to awaken motivation and interest in the studied subject, research for

VOLUME 8, 2020

developing quality professionals applying methodologies that leave behind the traditional method of decorating content and its mechanical reproduction. Thus, those researches have used different types of AM, as case studies involving the active learning of programming for students of different levels of education, since basic education up to postgraduate. Therefore, we can conclude that more research is required to improve innovative educational scenarios involving active learning in computer science.

Notably, the results shows that around 63% of the students (22 publications) analyzed applied AM as teaching strategies for teaching University. While around 20% of the studies (7 publications) were focused on Technical High School, just around 12% of the studies (4 publications) are related with others education levels, e.g., Postgraduate studies, High school, and Basic education. That scenario shows that AM are being adopted at all levels of education, i.e., it reveals that education professionals, especially in computing, are looking for innovate in relation to their teaching methodologies adopted in the classroom (see Fig. 4).



FIGURE 4. Education level system: High school, Postgraduate studies, Basic education, Not mentioned, Technical high school, University education.

Some results also revealed that the effects of the transition from a teaching-centered model of education to a learning-centered one involves a great cultural change for the University as an educational institution [43]. However, the social changes have leading to a change of perception in the teaching-learning process that promotes the emergence of the so-called active learning methodologies [18]. It is more perspective on higher education, so, we have analyzed the education level system in publications, because in higher education, there are growing trends toward flexible online course delivery [44].

Besides that, it is important to recognize that during the past few years, primary studies have been increased published regarding use AM as teaching strategies in the context of education [45]–[47] and have helped to refocus academic researches on method to teaching and learning, but it is still insufficient. Table 4 shows AM apply by education level system.

TABLE 3. Research goal per AM applied.

AM	Research Goal	Reference
ProjBL, PBL, TBL and PI	Increase students' motivation, belief in self-efficacy and autonomy, enhancing the learning of programming languages.	1
	Report the results of experience as the case study carried out for the first two periods of curriculum reformulation.	1
ProjBL and PBL	To present an analysis of AM aimed to learning undergraduate and graduate students.	1
GM, FC, and PI	Report experience of the use of gamification elements in conjunction with the hybrid teaching methodology in the inverted classroom modality applied to programming course.	1
FC	Resolution of practical problems, involving automation of solutions for monitoring network equipment and user actions, log analysis, data synchronization and automatic emission of alert messages, implemented in the Python programming language.	1
PBL	Shows how the use of PBL can contribute both to learning software engineering and to stimulating professional skills. Train programmers based on methodologies that focus on students learning and that favors the development of programming skills.	3
	Tests AM in undergraduate teaching and checks if students feel more engaged and motivated to carry out activities and develop their knowledge.	2
GM	Investigates the influence of gamification on students' engagement and performance in programming learning.	11
	Motivates students by serving as a tool to aid the teaching and learning process, making the practice of logic skills, and casting a light on creating algorithms, making fun and relaxed activity to be applied both inside and outside the classroom.	9
	Helps students to practice some introductory programming content in a fun way.	5

TABLE 4. AM applied by education level system.

AM	Education level system	Ref.
ProjBL, PBL, TBL and PI	Technical High School	1
ProjBL and PBL	University Education	1
	Postgraduate studies	1
GM, FC, and PI	University Education	1
FC	University Education	1
PBL	University Education	4
GM	Basic Education	1
	High School	1
	Technical High School	7
	University Education	17

We have observed that GM is being applied more among the methodologies mapped in this study, that its use ranges from basic education to graduation and it is present in 74% of the analyzed publications that use GM as an AM in teaching. Following, PBL methodology appears in 4 surveys, i.e., about 11%. Finally, the TBL, PI, FC and ProjBL methodologies are mentioned each in only 1 survey, i.e., about 2.80% of the analyzed publications.

The results presented above show us that, among the most significant and applied AM as teaching strategies in the computation, we can find GM and PBL. However, a few applications in the classroom were also using FC, PI, TBL and ProjBL. This scenario demonstrates that the use of those teaching methodologies, as teaching strategies in computing, can be seen as effective teaching methods, as they support the stimulation of the students' initiative to create opportunities for learning content inside and outside the classroom, because AM can promote proactivity, commitment to the educational process and linking learning to significant aspects of reality [48].

In additional, we also have analyzed the distribution of publications per year (Fig. 5). The interest of investigations



FIGURE 5. Publications per year. The figure includes line graphs showing number of publications returned by search and publications that met the inclusion criteria (2010-2019 period).

on topic of AM as teaching strategies in computing has began around 2010 with 1 publication. In the following years, from 2010 to 2015, publications remained stable, 1 or 2 paper per year. But, in 2016 and 2017 the number of publications has increased up to 4 and 7, respectively. 2018 confirms that evolution becoming the year in which most researches were publishing on that subject, 11 in total. But in 2019, that number fell again, just 4 publications.

However, we have considered an incipient number of publications for whom that applied of AM, supported by the principle of autonomy into the classroom, becomes of vital importance due to impact they can have on students' learning [49]. Also, AM have numerous forms of use for teaching and learning the computing courses. Furthermore, the applying of AM can favor the autonomy of the students both in face-to-face and distance education, favoring curiosity, stimulating individual and collective decision making, arising from activities of social practice and of students' contexts [50]. Therefore, AM emerges as a proposal to focus on the process of teaching and learning in the search for the



TABLE 5. Students' perceptions about AM.

AM	Students' Perception	Reference
ProjBL, PBL, TBL and PI	Developed confidence and interest in learning the contents.	1
ProjBL and PBL	Developed skills and experience in carrying out projects with execution efficiently.	1
GM, FC, and PI	Improve academic performance and, consequently, increasing programming skills.	1
FC	Represented a significant improvement in the group's performance.	1
PBL	Most students managed to learn the new concepts.	3
	Best comprehension and facilitated learning even for students inexperienced in programming.	2
	Focused on learning, being motivated to study, and learn the content provided. Students seem to have more desire to learn.	2
	Realized the importance of their own commitment, started to value the engagement of the team and collective practices.	1
GM	Learned basic programming concepts and developed logical-mathematical reasoning. There was a significant gain in learning the contents.	4
	Aroused great interest in the students remained engaged in their goals until the end of the activity.	2
	Stimulated by the new methodology, many students continued to attend the course.	3
	They had a pleasant experience and they managed to learn the proposed content.	2
	Students cited the game's ability to teach programming in a fun way.	2
	The students started to perform activities in the middle of the weekend, so they surprised, and this showed the degree of involvement provided by the game.	3
	Students assessed the gamified course more motivating than a non-gamified course and considered the practice interesting for learning.	4
	They wanted to learn more about the topic and noted that there was no difficulty in understanding. They were able to associate it with content already learned and that the level of complexity was adequate. The majority agreed that they learned summising or unexpected concepts.	2
	Participating in the proposed activities, students experience and carry out exercises that allowed them to verify their speed and practicality in performing them.	2
	Provided a positive experience for players regarding social interaction and fun. In addition, students pointed out that the experience with the game will contribute to their performance in professional life. Understanding about collaborative activity carried out by teams of students, considered the practice interesting for learning. It influenced them to study, so that the goals and activities could be fulfilled, which allowed them to advance in the game by their own effort.	3
	Learned and remembered the concepts of project management, because allowed students to fix the subjects better understanding the way of creating the missions, testing, and consequently improving the applicability of techniques.	2

active participation of all involved, centered on the reality in which they are inserted [51].

Thereby, that scenario shows combination of learning by challenges, the use of real problems and the games with the flipped classroom allow students to learn by doing, together and at their own place [41]. At the same time, teacher acts as an advisor, supervisor, and facilitator of the learning process, not only as the sole source of information and knowledge [45]. Thus, AM are teaching strategies centered on the effective participation of students in the construction of the learning process, in a flexible, interconnected and hybrid way [17].

B. STUDENTS' PERCEPTION ABOUT ACTIVE METHODOLOGIES IN TEACHING COMPUTING

RQ2. Which are the students' perceptions about the AM applied in teaching?

Table 5 shows AM applied and description in relation to students' perception; there were feeling of Satisfaction in learning the content, Motivation to learn content, and Feeling of mastery of content. Those results have indicated, in general, that the students' perceptions were positives regarding AM applied as strategy in teaching, i.e. AM enable greater interaction between teachers and students, benefiting both. Students acquire greater protagonism and independence in

VOLUME 8, 2020

teaching and learning process. Teachers have a great opportunity to innovate, to propose new ways to teach and to stimulate their students [42], but it is necessary their full commitment with all process. It is required from them to be aware of their role in order to achieve successful results, also need to actively practice the directions proposed by that type of methodology. So it can be seen as liberating for both in the sense of leaving the traditional forms of teaching and learning in favor of a new way to enhance the productive involvement of the teacher and the consequent active and innovative learning of the students.

The informations obtained from this mapping have made possible to identify the students' perceptions aroused with the use of different types of AM, enabling teachers to identify pedagogical practices that most attract student's participation. Because, reflecting on education in the contemporary context, includes, among other important aspects, the discussion on AM produced in a collaborative way and its implications for the experiences of students in the classroom [17].

We have considered that, from those three dimensions, the students' perspectives presented in some publications, in general, can support the construction of knowledge in order to monitor and to measure the data related to the acquisition and development of knowledge, skills and attitudes achieved by students, according to the research goals in question.

TABLE 6. Dimension of students' perception.

Dimensions in relation to students' perception	AM	Reference
Satisfaction in learning the	GM, ProjBL, PBL, FC,	23
content	PI and TBL	
Motivation to learn content	GM, ProjBL, PBL, FC,	22
	PI and TBL	
Feeling of mastery of	GM, ProjBL, PBL, PI	11
content	and TBL	

Also, many teachers imagine that all learning, including expository class, is inherently active. They consider that, while the students participates watching an expository class, they are actively involved, but cognitive science researches has indicated that students must do more than simply listen for an effective learning [18].

Therefore, from the adoption of active teaching methodological practices combined with Digital Information and Communication Technologies (TDICs), inherent to the CS, today's teacher and future teachers can make teaching and learning process more attractive to the eyes of students [52] and, for that, it is possible to use the AM as teaching strategies in computing new didactic methods seeking to combine the use of technologies, pedagogical knowledge and the different types of AM that currently exist for teaching.

Table 6 has mapping results organized in dimension identified to reply research question RQ 2. Not all articles explicitly have presented the students' perceptions regarding the applying of AM. Thus, some perceptions presented are based on the researchers' observations made during their field research. It can be observed when in some publications survey were used to learn about the acceptance and the applying of AM to learn the content. It was possible to identify three dimensions in relation to students' perception presented in those publications: a) Satisfaction in learning the proposed content: concerns the perception of the students about his learning of the concepts treated in the teaching of the subject's contents appears in 23 publication (65%); b) Motivation to learn the content: it is related to the students' perception of will and searches to learn even more about the concepts treated in the teaching of the subject's contents and it was observed in 22 publication (62%); and c) Feeling of mastery in relation to the proposed content: this perception is related to the ability to practice activities outside the classroom, to teach or share the content learned with colleagues and was notice in 11 papers (31%).

The new generation of students from the end of the 20th century seems not to be interested in attending a class in the same way every day [52]. Originally, the active method of learning works with the child's experience, so the teacher could support his students to reflect and encourage him to make decisions [53]. To understand the students' perceptions about the AM applied in teaching, we have applied different technical or studies to obtain the results presented.

TABLE 7. Technical or studies used per methodology.

Technical or Studies Used	AM	Reference
Online tool and survey	GM and PBL	6
Survey	GM	4
Survey and Game	GM	4
Computer tool	PBL and GM	2
Case study	PBL and GM	2
Case study and survey	PBL	2
Survey and framework	GM	1
Survey, online tool, and	GM	1
game		
Survey, online tool, and	FC	1
WhatsApp		
Survey and computer tool	GM	1
Case study and tool	PBL	1
Case study, survey, and	GM	1
game		
Framework	GM	1
Survey, framework, and	GM	1
online tool		
Game and interview	GM	1
Online tool and manual	GM	1
game		
Survey and manual game	GM	1
Survey and online course	ProjBL	1
Online tool	GM	1
Survey and game	GM	1
Survey, workshop, and focus group	GM	1

We can observe in mapping results that some researches have used more than one different technical or study to AM apply. Table 7 presents that scenario mapped.

It shows us that most of the researches have used the following technical or studies to get and analyze the perceptions of students: Survey, Online Tool, Games, Case Study, Framework, Manual Game, Computer Tool, Online Course, Focus Group, Interview, Workshop and Software. We have observed that most researches have used survey technical to gathering information about students' perspectives. In general, the goals was to explore the perception of students in the CS in order to analyze improvements, or the lack of it, in learning, according to different active teaching methodologies that have been used and in order to stimulate the knowledge skills of the studied contents.

Thus, we have noticed that survey, online tools, games, and case study were technical or studies most used by researchers. Also, those technical or studies were used together with others on most researchers analyzed, i.e. researchers as Oliveira & Barros [54] that applied the case study, game, and survey to get data about students' perceptions. In this sense, interview, focus group, workshop and online tools, online course, manual game, framework software and use of WhatsApp were combined with those technical or studies to understand the students' perceptions about the AM applied in teaching, presented in Table 7.

Thereby, the use of those techniques reflects the interest of the author to know how the students' perceptions about the AM applied in teaching of computing courses



FIGURE 6. Type of technical or studies used in research. Bars graph showing the number of technical that used to apply AM. In increasing order: Online course, Focus group, Interview, Software, WhatsApp, Manual game, Computer tool, Framework, Case study, Computer game, Online tool, and Survey.

could better support the learning process and could motivate researches to produce knowledge about process of teaching. Also, it becomes an important way to provide different and new ways of getting experiences in computing. So, in this context, it can be seen that most of the researchers have applied AM in teaching to instigate the students to participate in the class, through group works or discussion of problems. Those type of methodologies, as Lovato *et al.* [18] points, are those that place students as protagonists, while the teachers are mediators or facilitators of the process.

Fig. 6 shows some scenario about number of publications by type of technique or study used on the researchers analyzed. In relation of survey, this technical appears in 26 of publications. In sequence, the most used were online tools in 9 publications, the computer games in 7, and case study in 6 researches. The techniques or studies less used were the framework in 3, manual game and computer tool appears in 2 publications, while the others were, in general, used just one per each publication.

We observed that those techniques or studies have been used in education as a form to know different types of knowledge and to promote a critical, creative, and reflective attitude concerning to adoption of AM strategy in computer science. Because, the technological evolution that we are witnessing today brings challenges on the reality of education and so it points to need for changes that allows a redirection capable of answering those challenges and improve the skills of the new generations of students, since their educational beginning, allowing them to develop knowledge and skills for a future in which technology undergoes continuous developments [14].

Fig. 7 presents the distribution of publications per proposal presented by the research. We have noticed that between those proposals analyzed are commonly presented in works that seek to analyze students' school performance and as well as improve teaching and learning process. Also, those proposals seek to provide experience in the use of gamified on-line tools, to support teaching and learning in the most diverse course of computing area. They also seek to start the stimulation of computational thinking through a gamified



FIGURE 7. Proposals de studies mapping. Bars graph showing the type and number of proposals of studies that used to apply AM. In increasing order: Conceptual model, Extension course and a didactic guide, Distance programming course, Learning object, Application mobile, Game, Gamified environment educational, Gamified online tools, and Methodology.

application and to motivate students to practice some introductory content in the area of computing in a fun way.

Table 8 shows the mains courses worked on those research analyzed. It may reflect, in general, two propositions regarding traditional methodologies in computing: (i) those courses are more difficult to learn; and (ii) they are more difficult to teach using traditional methodologies. In any case, we can infer that those researchers were looking for solutions to problems that are evident, and we understand that this problem is not an exception. Therefore, the courses presented above can reflect the research' interest and necessity in knowing how AM can be used like teaching strategies centered on effective participation both the teacher and the students. Thus, on non-collaborative processes, teachers have more roles to perform, as work is controlled and organized by them, while in the collaborative process it is more opened and students become more active [55].

Thereby, it might help students to acquire autonomy over their interests and skills, motivating them to search and to research on the topics brought by contents. In short, they might realize that is worth learning [56], because with the appearance of the so-called AM of learning, the students becomes the protagonist and the use of this type of methodology allows the development of new skills, such as initiative, creativity, criticality reflective, capacity for self-assessment and cooperation to work as a team; and the teacher acts as an advisor, supervisor, and facilitator of the process [18], being necessary for this sense observing the different learning processes.

In this scenario, observing and evaluating the students' perception of his or her learning while AM are adopted as teaching strategies is not a trivial task. However, the literature shows that several studies are being carried out in order to analyze the potential of AM for learning [49], [52] or to facilitate retention of knowledge by students [18], [57]. Nevertheless, so far, there are few studies that aim to evaluate students' experiences in relation to active teaching methodologies [58]. Also, we have noticed that it is necessary, due to the similarities between some active methodologies, to give

TABLE 8.	Courses	worked	in ana	yzed	publications.
----------	---------	--------	--------	------	---------------

AM	Course	Reference
ProjBL, PBL, TBL and PI	Computer programming	1
ProiBL and PBL	Software testing	1
	Computer programming	1
GM, FC, and PI	Project management	1
	Computer programming	1
FC	Computer network	1
PBL	Programming logic	1
	Software engineering	1
	Troubleshooting I	1
	Computer programming	2
GM	Algorithms	2
	Data base	1
	Data structure	1
	Project management	1
	Introduction of	1
	informatics	
	Introduction of	2
	programming	4
	Programming logic	4
	Scientific methodology	1
	Computer programming	9
	Free and proprietary	1
	Software	1
	Theory of graphs	1
	Software testing	1

support for teachers on some type of learning. It is so because, in general, various AM were applied on those articles analyzed, however, in the course of this research, we have faced a lack of classification that could clearly present the use of AM in relation to the categorization of types of learning.

Thus, we have reached a conclusion that the interest in knowing the students' perception in this mapping of the literature can be very useful in different aspects regarding teaching and learning process. For example, it may be possible to know if some students need additional help to understand the information explained in the classroom or even to promote the intuition of means to help those students in their presented difficulty. Each study here analyzed has presented students' perceptions in different ways, i.e. they did not clearly show types of perception of the students regarding percentage of acceptance in relation to the applying of AM in the classroom. So, we think that is important present one classification that could help following researches and teachers that are interested on applying such methodology.

Therefore, considering the course of the computer science organized in three different fields, following the Reference Curriculum for Undergraduate Courses in Computing and Informatics by Brazilian Computer Society (SBC): I) Computability Fundaments (CF), which comprises the core of subjects that involve scientific parts and the fundamental techniques for the solid formation of the various computer program; II) Computing Technology (CT), which comprises the core of the course and represents a set of aggregated and consolidated knowledge that enable students to develop solving problems in the various application domains; and III) Information Systems (IS), which comprises the core of the course that enables students to use the resources of Information Technology in solving problems of productive sectors of society.

Table 9 presents the AM used per field of computing area in relation to the course worked on the researches analyzed in this paper. Therefore, it reflects, generally, that the GM and PBL were types of AM more used in three different area of computing. Thus, those types of AM can provide positives students' perception in relation of learning of course, because learning is most effective when it is active, experiential, situated, problem-based and provides immediate feedback [59]. Also, this research demonstrates that students engagement in course remains mixed, because, the GM and PBL were used in tree field of computing area and they are often put forth as a possible answer to change the teaching and learning landscape and to make it more attractive and interactive for students. Those types of AM, generally, can support students in relation of learning, because our results show that students' motivation and engagement is positively related to use of different AM in course.

TABLE 9. Computing area and active methodology applied.

Computing area	AM
Computability Fundaments	GM, PBL, ProjBL, TBL and PI
Computing Technology	GM, PBL and FC
Information Systems	GM, PI, ProjBL and PBL

ProjBL and PI are the types of AM most used to teach in the fields of IS and CF. It might means that teacher has to adapt his/her programming languages or programming technique, because the quality of the teaching is partly dependent on the teacher's competence in using the technology [12], and this type of course requires some changes in the teachers' teaching methods and different approaches in order to motivate and to engage students in program. ProjBL and PI are most used both in IS and CF. FC and TBL are most used only in CF. It might mean that teacher needs to actively learn how to work with different AM and has to get opportunities to try things out and to evaluates outcomes taking students' perceptions as basis, because courses of CF are core of CS learning.

So, to deal with the needs for more significant learning in CS, those types of AM can be developed and provide a richer engaging learning experience for teachers and students. Those methodologies can motivate processes of teaching and learning in different courses, because those methodologies might stimulate and engage all people involved in this process of teaching and learning. Also, it agrees with education literatures when they address AM like a set of processes, procedures, techniques, and tools that involve the students actively in the teaching and learning process.



TABLE 10. Benefits in relation to the adoption of AM.

AM	Description	Reference
ProjBL, PBL, TBL and PI	Improved students' confidence, encouraging them to seek answers to their questions on their own.	1
ProjBL and PBL	Integrated practice of different course. Development of high level social and cognitive skills.	1
GM, FC, and PI	Reconcile theory and practice and turn activities into meaningful learning.	1
	Significant increase in students' averages. The failure rate dropped. The approval rating has increased.	1
PBI	Significant improvement in group performance, significant increase in the average.	1
I BL	relation to their own involvement in teamwork and the involvement of other colleagues. Greater	2
	approximation between theory and practice.	
	It stimulated programming skills such as comprehension, analysis, observation of details, sequencing,	1
	and abstraction.	1
	The increased sense of responsibility of the students, who obtained engagement and course to study the	
	contents related to the context of the problem, better understanding the situation presented; encouraging	1
	reading, the use of logical reasoning and discussions; encouraging more detailed and in-depth	•
	investigation of the problem presented.	
	have a better understanding of the theories practices methods processes and tools that involve	
	Software Engineering Make classes dynamic and attractive: and created an environment of cooperation	
	between students. Encouragement of proactivity in the search for knowledge, confidence in your skills	1
	and that of your peers, commitment to solving tasks, teamwork, strategic thinking, and resilience in the	
	face of difficult tasks.	
GM	Make classes dynamic and attractive; and created an environment of cooperation between students,	
	serving as a stimulus for their integral development. Encouragement of proactivity in the search for	4
	knowledge, confidence in your skills and that of your peers, commitment to solving tasks, teamwork,	
	strategic thinking, and resilience in the face of difficult tasks.	
	Caused a decrease in any interaction resistance. It provoked greater interest in the students, when it was	
	role of a researcher involved in a rich narrative, thus simulating the activities that he should perform in	2
	his studies	
	Aroused creativity and the ability to think and act in a favorable condition in the conduct of computer	
	studies, making the learning process more attractive and interactive, understood as an important tool to	2
	support teachers in school education.	
	Considerable drop in the number of failing students. Improvement in students' performance in the	
	course. Minimized the students' difficulties in relation to the difficulty in learning computer	3
	programming. It motivates students' interaction, and engagement to carry out the activities.	
	Great impact on students' participation and development levels. Increase in attendance to classes.	2
	attention to support material and learning outcomes may benefit	2
	It brought together several elements and references, such as board, cards, characters, and obstacles, in a	
	simple system that adheres to the reality of the classrooms. The students developed five PC-related	6
	skills: data analysis skills, data collection skills, decomposition, abstraction, and data representation.	
	Motivated the study of computer programming at the levels: basic knowledge of programming and basic	
	concepts and assessment of the ability to read codes. Positive influenced on using gamification,	2
	engagement techniques to achieve the educational goals required to learn programming.	
	Competitiveness influences students' motivation to learn and overcome challenges. It provides the	
	students with learning in a playful and enjoyable way; makes it possible to relate scientific and abstract	2
	Concepts with familiar terms, with the students' reality.	2
	Reduction in the number of failures in the courses.	3

C. BENEFITS AND DIFFICULTIES IN RELATION TO ADOPTION ACTIVE METHODOLOGIES IN TEACHING OF COMPUTING

RQ3: Which are the benefits and difficulties in relation to the adoption of AM?

We have observed the most studies, in general, obtained benefits when they adopted GM as an AM for teaching, see Table 10. However, they also reflects difficulties (see Table 11), caused by the lack of commitment of students outside the classroom. Thereby, we have chosen to present those reports in relation to the benefits and difficulties presented in the texts of the publications. In general, it can support the use or adequacy of new teaching techniques applying AM, in addition to sharing the related benefits and difficulties in order to direct new practices. Attracting students' attention and keeping them involved are essential points for the process of learning and developing critical thinking. Teacher plays a role as an activator of learning [49]. Therefore, the adoption of this type of methodology for teaching in computing will certainly bring benefits for both students and teachers.

In the scenario presented by Table 10, we have understood that it is possible to infer that there is a positive influence on teaching and learning process in relation to the use of AM in CS courses. For example, GM, according to the analyzed works, is an engagement technique for achieve the educational goals required to learn programming. However, we also have observed that even with the positive influence of the use of AM as teaching strategies, there were yet difficulties (see Table 11), mainly in relation to the students' awareness

TABLE 11. Difficulty in relation to the adoption of AM.

AM	Description	Reference
ProjBL, PBL, TBL and PI	Lack of motivation and difficulties in understanding the content.	1
	Difficulty in motivating and mobilizing the attention of students. They can't obtain a broader view of the	1
	acquired concepts and establish a connection between theory and practice with greater effectiveness.	1
ProjBL and PBL	Not all students were able to acquire the skills in a timely manner to apply these concepts in solving the	
	problem. Some students, when finishing a problem without completing all the requirements, had	
	difficulties in continuing the same project in the subsequent problem. Finally, this change brought some	1
	difficulties in learning, the accumulation of deficiencies in projects of longer duration and the cognitive	
	overload in dealing with several concepts simultaneously.	
GM, FC, and PI	Lack of motivation of students to carry out practical activities. Lack of commitment on the part of some	
	students, who insist on not studying the content before the class (face-to-face meeting), thus promoting a	1
PC	lesser use in the course.	
FC	Leave the comfort zone, access the platform, and frequently participate in the flexible study environment	
	available. Make students aware of now to access and participate requently in the available materials, as	1
	students do not have the nabit of accessing the study station or actively participating in face-to-face	
	meetings. The source duration was considered insufficient for all the suggested activities to be completed	
	apprentished and the suggested activities to be completed	2
	control ability.	
	with basic questions about computer network theories. In autonomous studies, we observed difficulties	1
	for students to study about computer network incortes, in autonomous studies, we observed dimensions	1
PBL	Many academics have difficulties in novitioning themselves in front of colleagues and tutors to expose	2
TBE	their work and answer questions. Also some aspects related to the programming and operation of the	2
	computer and difficulties in relations to technical issues	
	Identify which students are really engaged in the tasks and which are collaborating with the team, since	
	simply sharing tasks does not represent group work.	1
GM	Low understanding of the basics in programine: a weak mathematical base: the lack of understanding of	
	the statements; exercises dissociated from "real" problems and little extra-class availability for studies.	6
	Difficulties in coding, due to the lack of programming logic base, presenting a great deficit of basic	
	technical knowledge necessary for the good progress of the course. Lack of basic prior knowledge to	4
	study data structure. Little time available for studies.	
	The lack of motivation of students and their difficulty in maintaining a continuous pace of study.	5
	Large classes, which make it impossible to carry out individualized monitoring and heterogeneous,	1
	which present disparity of knowledge and learning pace.	1
	Diversity of students learning rhythms, combined with large classes and lack of motivation.	1
	The workload required to keep the rankings and the site up to date, as two different platforms are being	1
	used to present and correct challenges.	1
	The need for faculty involvement and collaboration for the success of gamification and the right time for	1
	its application. In addition, achieving an appropriate balance between education and entertainment.	1
	Ensure the motivational factor of students. Mentality based on the repetition and memorization of	3
	concepts and content, without the concern of abstracting the knowledge acquired in solving problems.	2
	Lack of infrastructure and, in some cases, difficulties in applying non-traditional teaching methodologies	2
	due to students' cultural issues, awakening in teachers the need to use new technologies.	-

about importance of their self-studies organization and their autonomy, especially when adopted a technique in which teacher acts just presenting a problem and students should acting looking for solutions.

Table 11 shows difficulties presented by teachers. In general, those difficulties were related to use new teaching techniques. It is important to mention that those difficulties are not linked only to students' awareness of the importance of their studies, i. e., the difficulties involved, range from the structures available for teaching to the behavior of the teaching staff facing the new need to adopt different methodologies for more effective teaching. Thereby, it is necessary to know the method that can fulfill the needs of educational institutions in which the teacher operates, whether in basic education, high school, undergraduate and among others [60]. Because, within education, it is necessary to debate whether or not learning improves when students are allowed to explore the educational content by themselves or if students must be strictly guided in the topics to be learned. On the other hand, in classroom environments, teachers present students choices because they believe it increases effort and learning [61].

Those result have demonstrated that most researchers have found different benefits in relation to the AM. It is important to mention that not all publications clearly show the items benefits or difficulties. However, they present the results achieved and the difficulties faced in conducting their researches. Thereby, we can infer that the use of different AM positively influences the teaching because, in general way, it addresses different techniques for teaching,, and it addresses new activities and new engagement techniques required to achieve educational goals in learning different CS courses.

Therefore, AM have produced positive results for students in relation to learning and, for teachers, in relation to teaching practices, because, as Michaelsen have pointed [62], it forces students to break with their passivity. Considering the benefits presented in the analyzed publications, some advantages could be: (i) developing students' skills at a high level; (ii) promoting the development of personal and team skills; and (iii) also bringing advantages to the teacher, such as enthusiasm them in the classroom and career continuity as pointed by Guimarães *et al.* [52]. Those results are very important, because in the 20th century, education is a consequence of a process that involves several thinkers, who discuss teaching models and highlight the need for students' autonomy [18].

Finally, notably, all those researches also confirm a need for reflection about our traditional teaching practices and the adoption of AM applied in teaching and learning course on computing area. Because, according to the words of Vanbecelaere *et al.* [59] "learning is most effective when it is active, experiential, situated, problem-based and provides immediate feedback". So, without any doubts, technology has found its way into many classrooms around the world to help educational process to be more effective and enjoyable. Despite the great potential may technology have for facilitating and promoting students learning, teachers are also challenged to not only familiarize themselves with those technologies but to put it in practice as well [63].

V. THREATS TO VALIDITY

We believe that some limitations of this study can be related to publication, selection bias, inaccuracy in data extraction and erroneous classification. For Kitchenham & Charters [20], limitations related to publication bias refers to the problem that positive results are more likely to be published than negatives, because negative results take longer to be published or are mentioned in other publications to a lesser extent. In order to reduce that obstacle, as much as possible, we have done search on symposium, conference proceedings and journals in which the most relevant studies to the Computer and Education area are published frequently: Brazilian Symposium on Informatics in Education, Brazilian Symposium on Games and Digital, Entertainment, Computer Workshop at School, Computer Education Workshop, New Technologies in Education Magazine, Journal of Informatics in Education, International Congress of Educational Informatics and Scopus Library. This last platform is a digital library that is frequently used for researchers for access to relevant journals related to the same subject that we have been studying here. However, we did not consider other sources such as searching on conference bases or workshops that may affect, somehow, the validity of our results.

The bias that refers to the selection of publications is related to distortions that could happened in a statistical analysis due to the criteria used to select publications. In order to mitigate that threat, we have used the inclusion criteria to gather the largest possible number of publications that fit the AM as being applied in the teaching and learning activities of course of computing area, as well as the criteria for the exclusion of articles that did not present the necessary information related to this study. So, we have elaborated a SMS and validated it with other professionals of the areas with recognized experience in the conduction of that kind of work. We also have detected other limitations that could bring some problems to the results of this work and that are related to the systematic of procedure in carrying out this study. Thus, analyzing our main goals in relation to the accomplishment of this SMS, we have decided to categorize the selected VOLUME 8, 2020

publications and to identify representative studies instead of carrying out tests of validations on the results achieved by those authors. Besides that, we also have included on our mapping others specific questions, such as, regarding methodologies, technical or studies used, and this may have affected our results.

VI. CONCLUSION

This paper have presented a SMS that summarizes existing information regarding types of AM being applied in teaching and learning activities in CS courses. From an initial number of 753 papers, a total of 35 were selected for carrying out the mapping study. And the results obtained have allowed us to extract some conclusions regarding the state-of-the-art in the area, to identify several research gaps, and to extract some guidelines for innovative directions in computing education. Moreover, the application of a well-defined review protocol will also allow us to efficiently update and extend the SMS in future years.

As a result, our analyses have revealed or presented the following significant findings: a) an overview of the active methodologies applied to teach in computation that shows a variety of 6 types of AM; b) the AM used per field of computing area in relation to students' perception; c) different types of technical or studies used in research; d) and some benefits and difficulties in relation to the adoption of AM to teaching. According to those findings, we suggest that this review can greatly help and inform about the use of AM in the teaching and learning computing. Our review has showed a variety of AM that have been used and have identified the most common ones. It also has provided an overview of the methods used when validating the corresponding active methodology applied. In recent years, a great number of AM has been used as techniques for teaching in computing. However, in the current existing mapping studies, the perceptions of students have been forgotten. So, we have joined our mapping of both teachers' and students' points of view.

Finally, we expect that AM used in teaching computing in this review could also be effective in solving problems in other areas that share similar characteristics about learning difficulty, practice, and content abstraction students. Thus, it might allow teachers to experience the consequences of different methodologies choices as powerful approach to promote engagement, motivation, empathy, awareness, and constructive behavior for students. Also, we have highlighted the importance of continuing this research, such as those analyzed about mapping, in order to increase and to offer more consistent bibliography, both qualitative and quantitative data. That way, our research could sensibly allows the measurement and validation of the applicability of the AM as a strategic tool for teaching different contents in different course on computing or other areas.

REFERENCES

 R. M. Castro and E. S. Siqueira, "Alternative teaching techniques (active learning) for computer disciplines: A systematic mapping in the Brazil context," in *Proc. School Comput. Sci. Workshop*, 2019, vol. 25, no. 1, p. 1409.

- [2] P. Parsons, "Preparing computer science graduates for the 21st century," *Teach. Innov. Proj.*, vol. 1, no. 1, pp. 1–90, 2011.
- [3] M. Leon, "Quality teaching innovation and improvement of university teaching. Study and analysis project: Teaching innovation," Ministry Educ., Madrid, Spain, Tech. Rep., 2010, pp. 1–23.
- Educ., Madrid, Spain, Tech. Rep., 2010, pp. 1–23.
 [4] C. P. dos Santos and E. S. R. Soares, "Learning and teacher-student relationship at the university: Two sides of the same coin," *Study Aval. Educ.*, vol. 22, no. 49, pp. 353–369, 2011.
- [5] J. Moran, "Active methodologies and hybrid models in education," in New Digital Technologies: Reflections on Mediation, Learning and Development, S. Yaegashi et al., Eds. Curitiba, Brazil: CRV, 2017, pp. 23–35. Accessed: Jun. 27, 2020. [Online]. Available: http://www2.eca.usp. br/moran/wpcontent/uploads/2018/03/Metodológicas_Ativas.pdf
- [6] P. Demo, Teacher of the Future and Knowledge Reconstruction. London, U.K.: Voices, 2004.
- [7] M. A. de Armanda, R. L. Rodrigues, and E. V. C. Garcia, "A systematic mapping for problem based learning applied to computer science," in *Proc. School Comput. Sci. Workshop*, 2012, vol. 1, no. 1.
- [8] A. K. Hartwig, M. Silveira, L. Fronza, M. Mattos, and E. L. P. A. de Kohler, "Active methodologies for the teaching of computing: A systematic review and a practical study," in *Proc. Workshop Inform. School*, Nov. 2019, vol. 25, no. 1, p. 1, doi: 10.5753/cbie.wie.2019.1139.
- [9] J. V. Lima, M. de Melo Alves Júnior, A. Moya, R. Almeida, P. Anjos, M. Lencastre, R. A. de Araújo Fagundes Fagundes, and F. lencar, "Active methodologies and software engineering teaching: A systematic literature review," in *Proc. Workshop Inform. School*, Nov. 2019, vol. 25, no. 1, p. 1, doi: 10.5753/cbie.wie.2019.1014.
- [10] A. Raes, P. Vanneste, M. Pieters, I. Windey, W. Van Den Noortgate, and F. Depaepe, "Learning and instruction in the hybrid virtual classroom: An investigation of students' engagement and the effect of quizzes," *Comput. Edu.*, vol. 143, Jan. 2020, Art. no. 103682.
- [11] F. L. de Oliveira and F. E. I. Nunes, "Active learning in programming teaching: A systematic literature review," in *Proc. Workshops Brazilian Congr. Inform. Educ.*, 2019, vol. 8, no 1, p. 377.
- [12] M. A. de F. da Silva and E. M. Oliveira, "Educational robotics in the perspective of active methodologies," in *Proc. School Comput. Sci. Workshop*, 2019, vol. 25, no. 1, p. 1289.
- [13] B. Moreno, "Combining agile and active methodologies in teaching introduction to programming to high school students," in *Proc. Workshop Challenges Comput. Appl. Educ.*, 2019, vol. 8, no 1, p. 45.
- [14] R. R. Silva, J. Fernandes, and E. R. Santos, "Overview of the use of digital games in teaching programming at the higher education level in the last decade: A systematic literature review," in *Proc. Brazilian Symp. Comput. Educ.*, 2018, vol. 29, no 1, p. 535.
- [15] L. Bacich and E. J. Moran, Active Methodologies for Innovative Education: A Theoretical-Practical Approach. Coventry, U.K.: Penso Editor, 2018.
- [16] F. L. Lovato, A. Michelotti, and E. E. L. da Silva Loreto, "Active learning methodologies: A brief review," *Acta Sci.*, vol. 20, no. 2, pp. 154–171, 2018.
- [17] T. R. da Silva, T. Medeiros, H. Medeiros, R. Lopes, and E. Aranha, "Programming teaching-learning: A systematic literature review," *Brazilian J. Inform. School*, vol. 23, no. 1, p. 182, 2015.
- [18] B. Kitchenham and E. S. Charters, "Guidelines for performing systematic literature reviews in software engineering, version 2.3," Dept. EBSE, Keele Univ., Univ. Durham, Durham, U.K., Tech. Rep., 2007.
- [19] A. Fernandez, E. Insfran, and S. Abrahão, "Usability evaluation methods for the Web: A systematic mapping study," *Inf. Softw. Technol.*, vol. 53, no. 8, pp. 789–817, Aug. 2011.
- [20] J. Jesson, L. Matheson, and E F. M. Lacey, *Doing Your Literature Review: Traditional and systematic techniques*. Newbury Park, CA, USA: Sage, 2011.
- [21] T. Dyba, T. Dingsoyr, and E. G. K. Hanssen, "Applying systematic reviews to diverse study types: An experience report," in *Proc. 1st Int. Symp. Empirical Softw. Eng. Meas.*, Sep. 2007, pp. 225–234.
- [22] S. R. Lambert, "Do MOOCs contribute to student equity and social inclusion? A systematic review 2014–18," *Comput. Edu.*, vol. 145, Feb. 2020, Art. no. 103693.
- [23] K. Petersen, S. Vakkalanka, and L. Kuzniarz, "Guidelines for conducting systematic mapping studies in software engineering: An update," *Inf. Softw. Technol.*, vol. 64, pp. 1–18, Aug. 2015.
- [24] E. C. Moya, "Using active methodologies: The student-view," *Procedia-Social Behav. Sci.*, vol. 237, pp. 672–677, 2017, doi: 10.1016/j.sbspro.2017.02.040.
- [25] F. L. Noguero, Participatory Methodology in Education Universitaria, vol. 9. Madrid, Spain: Narcea Ediciones, 2005.

- [26] S. V. Silva and E. A. R. Gonçalves, "The simulated practice in teaching project management through the 'integrated management tool," Projects Manage., New Ideas Educ. Inform. TISE, Tech. Rep., 2015, p. 6.
- [27] R. A. Bittencourt, C. A. Rodrigues, and E. D. S. S. Cruz, "An integrated object-oriented programming experience, data structures and systems design with PBL," in *Proc. 21st Workshop Comput. Educ.-33rd Congr. Brazilian Computer Soc.* Maceió, Brazil: SBC, 2013, p. 10.
- [28] L. F. D. Pereira, F. Fábio Lapolli, F. F. Sampaio, C. L. R. Motta, and C. E. T. Oliveira, "Learning objects workshop: An approach to the teaching of computing in technical courses," *Brazilian J. Inform. School*, vol. 18, no. 3, pp. 4–18, Sep. 2010, doi: 10.5753/RBIE.2010.18.03.04.
- [29] D. Ravindranath, T. L. Gay, and M. B. Riba, "Trainees as teachers in team-based learning," *Academic Psychiatry*, vol. 34, no. 4, pp. 294–297, Jul. 2010.
- [30] R. H. Steadman, W. C. Coates, Y. M. Huang, R. Matevosian, B. R. Larmon, and L. McCullough, "Simulation-based training is superior to problembased learning for the acquisition of critical assessment and management skills," *Crit. Care Med.*, vol. 34, no. 1, pp. 151–157, 2006.
- [31] M. J. Lage, G. J. Platt, and M. Treglia, "Inverting the classroom: A gateway to creating an inclusive learning environment," *J. Econ. Edu.*, vol. 31, no. 1, pp. 30–43, Jan. 2000.
- [32] J. Bergmann and E A. Sams, *Flip Your Classroom: Reach Every Student in Every Class Every Day*. Washington, DC, USA: International Society for Technology in Education, USA, 2012, pp. 15–17.
 [33] R. Sayer, E. Marshman, and C. Singh, "Case study evaluating just-in-
- [33] R. Sayer, E. Marshman, and C. Singh, "Case study evaluating just-intime teaching and peer instruction using clickers in a quantum mechanics course," *Phys. Rev. Phys. Edu. Res.*, vol. 12, no. 2, Oct. 2016.
- [34] E. Mazur, Peer Instruction: The Active Learning Revolution. Coventry, U.K.: Penso Editor, 2015.
- [35] A. G. P. de Andrade, F. A. C. dos Santos, Jr., J. M. Pimentel, J. C. N. Bittencourt, and E. T. B. de Santana, "Application of the PBL method in software engineering teaching: Student's view," in *Proc. ERBASE*, SBC, Brazil, 2010, pp. 1–10.
- [36] J. W. Thomas, "A review of research on project-based learning," Autodesk Found., San Rafael, CA, USA, Tech. Rep., 2000, pp. 230–243.
- [37] J. R. Mergendoller and E. J. W. Thomas. (2001). Managing Project-Based Learning: Principles From the Field. Buck Institute for Education. [Online]. Available: httpwwwBieOrg
- [38] S. Deterding, R. Khaled, L. E. Nacke, and E. D. Dixon, "Gamification: Toward a definition," in *Proc. CHI Gamification Workshop*, Vancouver BC, Canada, vol. 12, 2011.
- [39] L. A. Ribeiro, T. L. da Silva, and E. A. Q. Mussi, "Gamification: A methodology to motivate engagement and participation in a higher education environment," *Int. J. Educ. Res.*, vol. 6, no. 4, pp. 249–264, 2018.
- [40] G. Kiryakova, N. Angelova, and E. L. Yordanova, "Gamification in education," in *Proc. 9th Int. Balkan Educ. Sci. Conf.*, 2014, pp. 1–5.
 [41] A. C. T. Klock, I. Gasparini, and E. M. S. Pimenta, "Designing, developing
- [41] A. C. T. Klock, I. Gasparini, and E. M. S. Pimenta, "Designing, developing and evaluating gamification: An overview and conceptual approach," in *Data Analytics Approaches in Educational Games and Gamification Systems*, A. Tilii and E M. Chang, Singapore: Springer, 2019, pp. 227–246.
- [42] J. Morán, "Changing education with active methodologies," *Coleç. Mídias Contemp. Media Convergences Educ. Cid. Youth Approaches*, vol. 2, no. 1, pp. 15–33, 2015.
- [43] A. Diesel, A. L. S. Baldez, and E. S. N. Martins, "The principles of active teaching methodologies: A theoretical approach," *Rev. Thema*, vol. 14, no. 1, pp. 268–288, 2017.
- [44] R. L. Moffitt, C. Padgett, and R. Grieve, "Accessibility and emotionality of online assessment feedback: Using emoticons to enhance student perceptions of marker competence and warmth," *Comput. Edu.*, vol. 143, Jan. 2020, Art. no. 103654.
- [45] E. F. Barbosa and E. D. G. de Moura, "Active learning methodologies in Professional and Technological Education," *Bol. Tec. Senac*, vol. 39, no. 2, p. 2, 2013.
- [46] R. Pereira, "Active method: Questioning techniques of reality applied to basic education and higher education," in *Proc. 6th Int. Colloq. 'Educ. Contemp.*, São Cristovão, Brazil, 2012, pp. 1–15.
- [47] D. Woods, Problem Based Learning: How to Get the Most From PBL. Hamilton, ON, Canada: McMaster Univ., 1994.
- [48] V. V. Lima, "Constructivist spiral: An active teaching-learning methodology," *Interface-Comun. Health Educ.*, vol. 21, pp. 421–434, Oct. 2016.
- [49] A. Corrêa, D. Zuasnabar, M. Santibanez, S. Silva, L. Prestes, and E. P. F. da Silva, "The use of mobile learning and active methodologies in the educational context," in *Proc. Workshops Brazilian Congr. Inform. Educ.*, 2019, vol. 8, no. 1, p. 730.
- [50] A. B. de Machado, "Active methodologies, integral knowledge," in *Innovative Practices in Active Methodologies*, S. R. Dias and A. N. Volpato, Eds. Florianópolis, Brazil: Context Digital, 2017.

- [51] D. B. Felippe, A. N. Volpato, I. Araldi, and E. S. R. Dias, *Innovative Practices in Active Methodologies*, S. R. Dias and A. N. Volpato, Eds. Florianópolis, Brazil: Context Digital, 2017, p. 179.
- [52] F. Guimarães, M. Leite, F. Reinaldo, and E. G. Ito, "Active teaching methods combined with technology for teaching practice: An experience report," in *Proc. Workshop Inform. School*, 2018, vol. 24, no. 1, p. 1, doi: 10.5753/cbie.wie.2018.333.
- [53] A. Prado, Understanding the 21st Century Student and How to Teach This New Generation. São Paulo, Brazil: Geekie, 2015.
- [54] E. Oliveira and R. Barros, "Gaia ABstração game: Proposal of a game to mediate the teaching-learning process of the object-oriented paradigm," in *Proc. 17th Symp. Bras. Games E Entertainment Digit.*, 2013, pp. 533–540.
- [55] P. L. Torres and E. A. F. Irala, "Collaborative learning," in *Some to Entertain Thinking and Act*, P. L. Torres, Ed. Curitiba, Brazil: SENAR, 2007, pp. 65–98.
- [56] J. H. Helm e L. G. Katz, Young Investigators: The Project Approach in the Early Years. New York, NY, USA: Teachers College Press, 2016.
- [57] L. de Almeida, M. P. Rolim, R. da Silva, and E. A. Costa, "E-PBL?: Support tool for learning and using the problem based learning methodology," in *Proc. Workshop Inform. School*, Nov. 2019, vol. 25, no. 1, p. 1, doi: 10.5753/cbie.wie.2019.1399.
- [58] J. B. S. Júnior, A. Kronbauer, and E. J. Campos, "Assessing users' experience with active methodologies in Human-Computer Interaction classes," in *Proc. Brazilian Symp. Inform. School*, Nov. 2019, vol. 30, no. 1, p. 1, doi: 10.5753/cbie.sbie.2019.1201.
- [59] S. Vanbecelaere, K. Van den Berghe, F. Cornillie, D. Sasanguie, B. Reynvoet, and F. Depaepe, "The effects of two digital educational games on cognitive and non-cognitive math and reading outcomes," *Comput. Edu.*, vol. 143, Jan. 2020, Art. no. 103680.
- [60] T. S. Borges and G. Alencar, "Active methodologies in the promotion of critical student education: The use of active methodologies as a didactic resource in the critical education of higher education students," *Cairu Rev.*, vol. 3, no. 4, pp. 119–143, 2014.
- [61] C. Ferguson, E. L. van den Broek, and H. van Oostendorp, "On the role of interaction mode and story structure in virtual reality serious games," *Comput. Edu.*, vol. 143, Jan. 2020, Art. no. 103671.
- [62] L. K. Michaelsen, "Getting started with team learning," in *Team Learn. Transform. Use Small Groups Westport.* Westport, CT, USA: Greenwood, 2002.
- [63] R. Scherer, F. Siddiq, and J. Tondeur, "All the same or different? Revisiting measures of teachers' technology acceptance," *Comput. Edu.*, vol. 143, Jan. 2020, Art. no. 103656.
- [64] T. S. C. da Silva, J. C. B. de Melo, and E. P. C. A. R. de Tedesco, "A model to promote student engagement in programming learning using gamification," *Brazilian J. Inform. School*, vol. 26, no. 3, p. 3, 2018.
- [65] E. Pantaleão, L. R. Amaral, and E. G. B. E. Silva, "An approach based on the Robocode environment for teaching high school programming," *Brazilian J. Inform. School*, vol. 25, no. 3, p. 3, 2017, doi: 10.5753/rbie.2017.25.03.95.
- [66] M. C. Cera, M. H. D. Forno, and E. V. G. Vieira, "A Proposal for Teaching Software Engineering from Problem Solving," *Brazilian J. Inform. School*, vol. 20, no. 3, p. 3, Dec. 2012, doi: 10.5753/rbie.2012.20.03.116.
- [67] S. V. Silva and A. R. Gonçalves, "The simulated practice in teaching project management through the integrated management tool," Project Manage., New Ideas Educ. Inform. TISE, Tech. Rep., 2015, vol. 6.
- [68] A. F. F. Costa, A. F. M. F. de Melo, G. G. Moreira, M. A. de Carvalho, and E. M. V. A. de Lima, "Application of inverted room and gamification elements to improve teaching-learning in object oriented programming," Project Manage., New Ideas Educ. Inform. TISE, Tech. Rep., 2017, vol. 13, pp. 223–232.
- [69] W. Nagai, C. Izeki, and E. R. Dias, *Experience in the Use of Gamified Online Tools in the Introduction to Computer Programming*, vol. 22, no. 1, 2016, p. 301.
- [70] M. G. de Oliveira, A. Neves, M. F. S. Lopes, H. F. Medeiros, M. B. Andrade, and E. L. L. Reblin, "A distance programming course with active methodologies and learning analysis by software metrics," *Rev. New Technol. Educ.*, vol. 15, no. 1, pp. 1–10, 2017.
- [71] R. I. Casarotto, G. Bernardi, A. Z. Cordenonsi, and E. R. D. Medina, "Logirunner: A board game as a tool to aid the teaching and learning of algorithms and programming logic," *RENOTE-Rev. News Tecnol. Educ.*, vol. 16, no. 1, pp. 1–10, 2018.
- [72] R. I. Casarotto, G. Bernardi, A. Z. Cordenonsi, and R. D. Medina, "The use of unplugged computing in a gamification context for teaching data structures," *Renote-Rev. News Tecnol. Educ.*, vol. 16, no. 2, pp. 546–555, 2018.

- [73] J. A. Moreira and W. M. Monteiro, "Inverted classroom—An experience in teaching-learning programming for computer network administration," *Rev. New Technol. Educ.*, vol. 16, no. 1, pp. 1–11, 2018.
- [74] M. E. C. Natal, B. A. Barbosa, J. C. Hernandes, B. de Sousa Much, M. Bigolin, S. J. R. da Silva, C. B. Silva, and L. F. B. de Carvalho, "Trilogic: A gamified environment as an aid tool for teaching programming logic learning," *Rev. New Technol. Educ.*, vol. 16, no. 2, pp. 41–50, 2018.
- [75] D. S. de Souza Rabelo *et al.*, "Development of computer systems using problem-based learning," in *Proc. Brazilian Symp. Comput. Educ.*, 2018, vol. 29, no. 1, pp. 188–197, doi: 10.5753/cbie.sbie.2018.188.
- [76] F. I. R. Pessoa, A. L. S. O. Araujo, W. Andrade, and E. D. Guerrero, "T-mind: A gamified application to encourage the development of computational thinking skills," in *Proc. Brazilian Symp. Comput. Educ.*, 2017, vol. 28, no. 1, p. 645.
- [77] M. A. de Azevêdo Silva and A. Dantas, *KLouro: An Educational Game to Motivate Students Who Are New to Programming*, vol. 25, no. 1, 2014, p. 702.
- [78] E. H. S. Raposo and V. Dantas, "The snake challenge-using gamification to motivate students in an introductory programming discipline," in *Proc. Brazilian Symp. Comput. Educ.*, 2016, vol. 27, no. 1, p. 577.
- [79] M. C. Meireles and B. Bonifácio, "Use of agile methods and problembased learning in software engineering teaching: An experience report," in *Proc. Brazilian Symp. Comput. Educ.*, 2015, vol. 26, no. 1, p. 180.
- [80] R. F. de Azevedo and B. C. de Paula, "Proposed methodology for learning computer programming through the recontextualization of serious games in the style game & watch," in *Proc. SBC SBGames*, 2011, pp. 1–9.
- [81] R. T. Figueiredo and C. Figueiredo, "Wargrafos- game to aid in learning the discipline of graph theory," in *Proc. 12th Symp. Bras. Games Entertainment Digit. SBGames*, 2011, pp. 1–9.
- [82] E. D. de Oliveira and R. M. de Barros, "Gaia ABstração game: Proposal for a game to mediate the teaching-learning process of the object-oriented paradigm," in *Proc. 17th Brazilian Symp. Games Digit. Entertainment*, São Paulo, Brazil, 2013, pp. 533–540.
- [83] S. A. Melo, "Game of code: Applying gamification in programming disciplines," in *Proc. SBGames XV SBGames*, São Paulo, Brazil, Sep. 2016, pp. 1241–1244.
- [84] J. Mattar *et al.*, "Gamification and games for scientific methodology: Proposed board game and game," in *Proc. SBGames XVI SBGames*, Curitiba, Brazil, Nov. 2017, pp. 757–763.
- [85] S. Melo and C. d S. S. Neto, "Game of code: Development and evaluation of a gamified activity for programming disciplines," presented at the em 16th Symp. Brazilian Games Entertainment Digit. (SBGames), 2017.
- [86] E. D. de Oliveira, L. de Souza Mendes, M. C. Camargo, M. H. S. Senegalha, and R. M. de Barros, "Gaia abstraction game BD: A game that aims to help the teaching-learning process of entify-relationship concepts and relational model," in *Proc. SBGames, XVI SBGames*, Curitiba, Brazil, Nov. 2017, pp. 27–33.
- [87] J. Mombach et al., "POOkemon: A game about object-oriented programming," in Proc. SBGames, 2018, p. 5.





rently pursuing the Ph.D. degree in informatics with the Federal University of Amazonas (UFAM). Her research interests include software engineering education, active learning strategies, and related topics. She is currently a Professor of basic, technical and technological education with the Federal Institute of Education, Science and Technology of Rondônia (IFRO)/Campus Porto Velho North Zone.

MARIA IVANILSE CALDERON RIBEIRO is cur-

ODETTE MESTRINHO PASSOS received the Licentiate and bachelor's degrees in mathematics, the master's degree in informatics, and the Ph.D. degree in informatics from the Federal University of Amazonas, in 1995, 2003, and 2014, respectively. She is currently an adjunct Professor with the Federal University of Amazonas / Institute of Exact Sciences and Technology. She has experience in computer science, working mainly in informatics in education, software pro-

cess improvement, and experimental software engineering.